Author's response to reviews

Title: Effects of Hypericum Perforatum, in a rodent model of periodontitis.

Authors:

Irene Paterniti (ipaterniti@unime.it)
Erico Briguglio (enricobrigu@hotmail.com)
Emanuela Mazzon (ehazzon@unime.it)
Maria Galuppo (mariagaluppo@gmail.com)
Giacomo Oteri (oteri@unime.it)
Giancarlo Cordasco (gcordasco@unime.it)
Salvatore Cuzzocrea (salvator@unime.it)

Version: 4 Date: 15 November 2010

Author's response to reviews:

November 15, 2010

Editor-in-Chief:,
BMC complementary and alternative medicine
MS 1538219724619619

Dear Editor

Please find enclosed the revised version of the above-referenced manuscript. According to the suggestions we rearrange the manuscript.

I would like to thank the Editor and the Referee for their skillful comments that improved the quality of our paper.

Sincerely yours,

Salvatore Cuzzocrea Ph.D.

Address: Institute of Pharmacology, School of Medicine, University of Messina, Policlinico Universitario, Torre Biologica 98123 Messina Italy.
Tel.: (39) 090-2213644
Fax (39) 090-2213300
email: salvator@unime.it